Author's response to reviews

Title: The effectiveness of a semi-tailored facilitator-based intervention to optimise chronic care management in general practice: a stepped-wedge randomised controlled trial

Authors:

Tina Drud TDD Due (tina.due@sund.ku.dk)
Thorkil TT Thorsen (tthorsen@sund.ku.dk)
Marius Brostrøm MBK Kousgaard (marbro@sund.ku.dk)
Volkert Dirk VDS Siersma (siersma@sund.ku.dk)
Frans Boch FBW Waldorff (fransw@sund.ku.dk)

Version: 5 Date: 2 April 2014

Author's response to reviews:

Dear Magdalena Morawska

Thank you for accepting your manuscript.

I have made the required changes and gone through manuscript formatting checklist. This has resulted in minor changes in both the manuscript and the figures. The revised tables are now in the manuscript and a new version of the manuscript has been uploaded. New versions of the three figures are uploaded as well.

Regards Tina Due